Author's response to reviews

Title: Myasthenia gravis and pregnancy: Clinical implications and neonatal Outcome

Authors:

Jose Francisco Tellez-Zenteno (jftellez@quetzal.innsz.mx)
Lizbeth Hernandez-Ronquillo (liztellez33@hotmail.com)
Vicente Salinas (vicente_salinas@latinmail.com)
Guillermo Garcia Ramos (garcira@quetzal.innsz.mx)
Bruno Estanol (bestanol@hotmail.com)

Version: 2 Date: 3 July 2004

Author's response to reviews: see over
Thank you for the commentaries.

Responses

1.-We have added the numbers of the newborns

2.-The article has been reviewed for an expert in the English language.

3.-We corrected the mistakes in the tables and the text.

4.-We deleted the table 3; we included all the information in the discussion.